Author's response to reviews

Title: Favourable ten-year overall survival in a caucasian population with high probability of hereditary breast cancer

Authors:

Laura Cortesi (hbc@unimo.it)
Cristina Masini (cristinamas@libero.it)
Claudia Cirilli (cirilli.claudia@policlinico.mo.it)
Veronica Medici (veronica.medici@gmail.com)
Isabella Marchi (robertoisa@alice.it)
Giovanna Cavazzini (giovanna.cavazzini@ospedalimantova.it)
Giuseppe Pasini (Giuseppe.Pasini@ausln.net)
Daniela Turchetti (danielat@med.unibo.it)
Massimo Federico (mfederico@unimore.it)

Version: 6 Date: 3 March 2010

Author's response to reviews:

Dear Editor,

formatting changes have been made:

Tables: tables with shading and colours have been substitute with bold characters and the vertical lines have been removed.

References: Reference 15 is now cited in square brackets on page 9.

Title: we have capitalised only the first word without full stops/periods. Affiliations: we have added the city in the affiliation details.

Email address: we have included email addresses for all the authors on the title page.

Introduction: we have renamed this section 'Background'.

Patients and methods: we have renamed this section 'Methods'.

Authors' contributions: we have formatted the section into a paragraph

Figures: we have minimised white space around the image.

Figures: we have erased the title (Fig.1 etc...)